Author's response to reviews

Title: Do decision support systems influence variation in prescription?

Authors:

Judith D de Jong (j.dejong@nivel.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)
Peter Spreeuwenberg (p.spreeuwenberg@nivel.nl)
Gert P Westert (gert.westert@rivm.nl)
Dinny H de Bakker (d.debakker@nivel.nl)

Version: 5  Date: 29 January 2009

Author's response to reviews:

Dear editors,

We are pleased that our manuscript is accepted for publication.

We made the formatting changes that were requested:

• All references are cited in the main body of the manuscript.
• Appendices are numbered using Arabic numbers.
• The unnecessary line break on the title page has been removed.
• Table titles are now one sentence long.
• Appendices have been placed before the References section.

Kind regards, on behalf of all authors,
Judith D. de Jong